
-----------------------------------------------------------------------------
                Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
        Wed Dec  1 18:05:42 GMT 2010
Year: 10  Doy: 335
Observer: koon
WEATHER COMMENT: Wed Dec  1 18:05:47 GMT 2010
Foggy, wind=15mph from the SE, temp=34F.
___end___
H-ALPHA COMMENT BY DARRYL: Wed Dec  1 20:08:59 GMT 2010
I moved the PC to be above the Nene Sparc, this frees up room in the lower
cabinet for the HawTel network upgrade equipment.
___end___
WEATHER COMMENT: Wed Dec  1 23:00:36 GMT 2010
Still have clouds too thick for spar to track.
___end___
NOTE BY DARRYL: Wed Dec  1 23:00:58 GMT 2010
All day I’ve been working on preparing a place for the the "in-MLSO" HawTel
equipment needed for the network upgrade.  Today Mac and Oly from HawTel came up
and installed almost all the equipment, they need to come back, possibly with
Norman for the "turn up" after the installation, or possibly before to install
a few more cards in the equipment and install at least one cable that will
go to the panel on the inside of the North wall that the Cable/conduit crew
will install next.  Also another "Splicing" crew will need to come up during
the installation process.  Note that Norman will be here with Lance and us 
for the "turn up".
___end___
GONG COMMENT BY DARRYL: Thu Dec  2 00:22:10 GMT 2010
I went to shelter and swapped the Halfwave Amp, waveplate card, and Drive ISO card.
I left SN info entered as a comment at the shelter workstation, restarted obs
and it is OK.  This was requested by Bert.
___end___
NOTE BY DARRYL: Thu Dec  2 00:24:45 GMT 2010
Rebooted Kenobi as requested by Alice and Vic last week.
___end___
Thu Dec  2 01:00:45 GMT 2010 
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